Author's response to reviews

Title: High prevalence of Treponena pallidum infection compared to HIV, HBV and HCV infections and transfusion complications among blood donors and recipients at the Laquintinie hospital in Douala, Cameroon

Authors:

Carole E Eboumbou Moukoko (elsemoukoko@yahoo.fr)
Françoise Ngo Sack (nsackmakak@yahoo.fr)
Estelle G Essangui Same (essanguestelle@yahoo.fr)
Madeleine Mbangue (madombangue@yahoo.fr)
Léopold G Lehman (lblehman@yahoo.fr)

Version: 2 Date: 13 August 2013

Author's response to reviews:

Two modifications have been done:
- The contribution of Pr. Lehman as you requested.
- The order of the first names of the first author:
  • Before: Else Carole Eboumbou Moukoko and Abbreviation was ECEM
  • Now: Carole Else Eboumbou Moukoko and Abbreviation is now CEEM